Author’s response to reviews

Title: Home based records for poor mothers and children in Afghanistan, a cross sectional population based study

Authors:

Sayeed Saeedzai (atasayedzai@gmail.com)
Iftekhar Sadaat (iftekharsadaat@gmail.com)
Zelaikha Anwari (zanwari.mophrhd@gmail.com)
Yohei Ishiguro (Ishiguro.Yohei@jica.go.jp)
Shafiqullah Hemat (shafiqullahhemat8@gmail.com)
Shakir Hadad (hadad.rhd@gmail.com)
Keiko Osaki (Osaki.Keiko@jica.go.jp)
Megumi Asaba (masaba@tulane.edu)
Rasuli Mudassir (RasuliAhadmuddassir.AN@jica.go.jp)
Jane Burke (Machlin.burke@gmail.com)
Ariel Higgins-Steele (ahiggins@unicef.org)
Khaksar Yousufi (kyousufi@unicef.org)
Karen Edmond (karen.edmond@uwa.edu.au)

Version: 1 Date: 23 Nov 2018

Author’s response to reviews:

Editor comment - In your “ethical approval and consent to participate” section of your declarations please state if written informed consent to participate was obtained from the parents/guardians of the minors included in this study (minors are considered anyone under the age of 16).

Response - This change has now been made in the declarations section as requested, and the revised manuscript has been uploaded